CSE 326: Data Structures Spring 1999 Course Information

Instructor: Anna Karlin Sieg 426C karlin@cs.washington.edu 543-9344

TA: Jason Hartline Sieg 226A hartline@cs.washington.edu

Office hours: Anna: Friday, 1:30 – 2:30pm and by appointment, in Sieg 426C.

Jason: Monday, 12:30 – 1:20, Thursday, 1:30 – 2:30, and by appointment, in Sieg 226A.

Prerequisites: CSE 143, CSE 321.

Grading: Homework (40%), Midterm (20%), Final (40%). Percentages are approximate. The final exam is scheduled for Wednesday, June 9, 1999, 2:30-4:20 p.m. in our normal lecture room.

Computing facilities: Sieg 329 Unix lab. Programming assignments will be done in C++.

Course Web: All handouts and a log of all mail sent to the class mailing list will also be available on the course web at the following URL:

http://www.cs.washington.edu/education/courses/326/CurrentQtr/

Mailing List: The class mailing list is cse326@cs.washington.edu. Instructions for subscribing to the class mailing list can be found on the course web.

We will use this list for announcements of general interest to the class. Students should also feel free to use it to ask questions, post information, or initiate discussions of general interest to the class. Questions or comments that are not of general interest should instead be directed to the TA or instructor at the addresses above.

Textbook: Mark Allen Weiss, Data Structures & Algorithm Analysis in C++, Addison Wesley, Second Edition.

Reading assignment: Chapters 1– 3. Material primarily from Chapter 2 will be covered in the first two lectures

If you would like to request academic accommodations due to a disability, please contact Disabled Student Services, 448 Schmitz, 543–8924 (V/TDD). If you already have a letter from Disabled Student Services indicating you have a disability that requires academic accommodations, please let the instructor know so we can discuss the appropriate accommodations.